



Article Alliances between For-Profit and Non-Profit Organizations as an Instrument to Implement the Economy for the Common Good

Carmen Talavera and Joan R. Sanchis *D

Business Administration Department, Faculty of Economics, University of València, Tarongers Av., 46022 València, Spain; tacar@alumni.uv.es

* Correspondence: joan.r.sanchis@uv.es

Received: 29 October 2020; Accepted: 13 November 2020; Published: 15 November 2020



Abstract: The model of the Economy for the Common Good (ECG) has cooperation as one of its main principles. This alternative economic model proposes to prioritize cooperation over competition to favor the creation of social value. From this point of view, strategic alliances between organizations can be used as an instrument that supports implementation of the ECG model. In recent years, alliances between for-profit and non-profit entities have been strengthened as a method to facilitate actions focused on social responsibility and sustainability. Moreover, the ECG model has become an adequate management framework for corporate sustainability. This work aims to connect alliances between for-profit and non-profit organizations with the ECG model. First, this connection is manifested in a theoretical way. This paper is going to analyze how such alliances can contribute to increasing the values of the ECG model: human dignity, solidarity and social justice, environmental sustainability, and transparency and codetermination. Afterwards, this work analyzes two cases of this type of alliance—Grupo Vips-Fundación Hazlo Posible and Danone Foods-Grameen Bank—to determine the benefits that this type of cooperation can provide to society. We study their motives and the benefits that they bring to the organizations and the community. Therefore, this work assesses how these types of alliances influence the different topics included in the Common Good Matrix. Moreover, we conduct a comparative analysis between both cases. This work demonstrates that, by implementing this type of strategic alliances, the creation of social value is favored, thus contributing to implementation of the ECG model.

Keywords: economy for the common good; cooperation; strategic alliances; sustainability; shared value; non-profit organizations

1. Introduction

Cooperation between organizations is a practice for which history dates back to the beginning of economic exchanges. Nevertheless, it becomes more relevant when the environment starts to be considered a significant variable in strategic analysis during the second half of the twentieth century. The economic dynamism and the globalization of markets restrict an organization's capacity to respond individually, an aspect that forces them to establish cooperation agreements to face environmental challenges. In consequence, alliances become a critical strategic tool for business success [1]. The literature on this subject is extensive [2], and the names used to refer to it are diverse: cooperation agreements [3,4], alliances [5,6], or coalitions [7], among others. Furthermore, alliances are analyzed from multiple perspectives and approaches [8]. Most authors focus on aspects such as their motives [9], the process of formation of the agreement [10], or their management [1].

Nowadays, a more recent source of research analyzes alliances and their relationship with ethical behaviors, social responsibility, and sustainability in corporations [11,12]. These studies affirm that the success of cooperation agreements depends on the absence of opportunistic behaviors, mutual respect, and trust. These fundamental aspects for the management of alliances connect with the social values derived from social responsibility and business ethics. Therefore, alliances between for-profit and non-profit entities are being promoted in order to develop critical competencies to create social value [13–17].

Furthermore, there is another aspect which is currently generating a broader literature on the subject of corporate sustainability. Such an aspect is the relevance of social value creation and its relationship with economic value through what Porter and Kramer [18] refer to as shared value [19–24]. Dyllick and Muff [25] establish a matrix in which they explain the process through which a company can reach "true sustainability". The process allows for identification of a typology formed by four different stages: (1) business-as-usual (economic concerns), (2) business sustainability 1.0 (three-dimensional concerns), (3) business sustainability 2.0 (triple bottom line), and (4) business sustainability 3.0 (outside-in). These authors point out that the Economy for the Common Good (ECG) model is an organizational framework that focuses on business sustainability 3.0 [25] (p. 169).

The ECG model is an organizational framework created in 2010 in Europe (Austria) by economics professor and social activist Christian Felber [26]. According to this model, corporations can measure their contribution to the common good through the Balance Sheet for the Common Good. This tool uses a strategic matrix—the Matrix for the Common Good—to quantify the company contribution to the creation of social and environmental value. This contribution is related to four principal values (human dignity, solidarity, social justice, environmental sustainability, and codetermination and transparency) with five different stakeholders (suppliers, investors, employees, customers, and social environment) [27]. This model replaces profit for the common good, and it gives priority to cooperation over competition. In this sense, the ECG is an organizational model that highlights the cooperation between companies, both internally (intra-cooperation) and externally (inter-cooperation).

When analyzing the literature on strategic alliances, we can find a few works concerning the importance of strategic alliances between for-profit and non-profit organizations. From an empirical point of view, the literature is even scarcer. In this sense, the present work provides new literature on this subject. Furthermore, we intend to relate this type of cooperation agreement with sustainability models such as the ECG one to prove that such alliances can help implement this model. Once the work establishes the theoretical relationship between both, future investigations could quantify the contributions of these alliances to the ECG model by applying the Common Good Matrix.

The objective of this work is to relate the study of alliances with the ECG model to establish a theoretical connection between the two. To this end, this work analyzes empirically the contribution that alliances between for-profit and non-profit organizations can make to the ECG model.

Cooperation agreements between these two types of organizations contribute to the creation of social value, which is the principal purpose of the ECG model. Therefore, establishment of these types of alliances can contribute to implementation of the ECG model in companies. This work intends to establish the theoretical connection between both approaches and the empirical contributions that these types of alliances can make to the Common Good Matrix.

With this purpose, our work is structured as follows. After this introductory section, we introduce Section 2, which includes the literature review. We divide such segments into three different subsections: (1) "2.1. Cooperation between for-profit and non-profit organizations", (2) "2.2. The Economy for the Common Good organizational model", and (3) "2.3. Cooperation in the ECG model". Following this segment, we describe the work method in Section 3. Afterwards, Section 4 focuses on the description of case studies. This segment is divided into two different parts: (1) "4.1 Grupo Vips–Fundación Hazlo Posible" and (2) "4.2. Danone Foods–Grameen Bank". After this, Section 5 analyzes the results of the empirical study conducted previously, employing a comparative analysis between both cases. To finish, Section 6 discusses the results of this work, and Section 7 contains the conclusions.

2. Theoretical Framework

2.1. Cooperation between For-Profit and Non-Profit Organizations

Interfirm cooperation as a field of academic study within the economy began to gain relevance in the 1970s. Since that time, numerous papers have highlighted the importance of cooperation to obtain a competitive advantage [28–30]. The first papers on this subject affirmed that cooperation was a cover-up practice that restrained competition [31–36]. However, in the 1980s, the cooperation approach experienced a change due to application of the "Transaction Costs Economics" [37,38]. Nowadays, cooperation is analyzed as a hybrid or intermediate form between the market and the company. Interfirm cooperation reduces market inefficiencies (reducing transaction costs), and it facilitates access to resources and capabilities that the entities could not obtain individually [39,40].

From an organizational perspective, cooperation begins to gain interest as a field of academic study when the environment starts to be considered a critical element in the competition between organizations [41]. Even though there are previous references [42], the most relevant works emerge when cooperation is studied within the resource dependency approach. Such an approach states that cooperation between entities occurs as a consequence of mutual dependence [43,44]. This fact leads to a strategic approach to cooperation, which is considered an alternative strategy to mergers and vertical integration (internalization of activities). This approach comprises two main types of reasons why companies cooperate: (1) to access new resources and capabilities that the entities do not possess [45,46] and (2) to organize certain activities of the value chain [7,47,48].

The existing literature indicates that cooperation can be implemented through different methods—functional agreements, franchising, outsourcing, joint ventures, networks, etc.—which share common characteristics that help define the concept [28]. Such attributes are as follows: (1) partners keeping their legal independence, (2) the explicit and long-term nature of the collaboration, (3) partners sharing common and compatible goals, and (4) the organizations involved sharing their resources and capabilities.

The strategic perspective of organizations focuses the study of alliances on two different aspects [7,9,47,49]: (1) their motives and causes, and (2) their consequences (benefits and results of the alliances). Moreover, this perspective includes the study of the process of negotiating and managing the cooperation agreements [50–52].

With this work, we are interested in introducing a new motive that has been little analyzed by the academic field: the use of alliances to promote the implementation of socially responsible practices, ethical behaviors, and sustainability criteria [11,53,54]. Table 1 summarizes the main contributions concerning the relationship between strategic alliances and corporate social responsibility.

Authors	Contributions
Canzaniello et al. (2017) [30]	This paper analyzes the vertical alliances between companies and their suppliers in the supply chain under sustainability criteria.
Altamira (2000) [55]	This paper studies how ethical behaviors on alliances can avoid opportunistic behaviors and distrust between partners.
Michavilla (2011) [56]	This work states that the success of an alliance depends on mutual trust between partners.
Renart Cava (1999) [57]	The author concludes that the failure of an alliance depends on unethical behaviors and the loss of trust between partners.
Pérez López (1993) [58]	This paper states that the transcendent motivations are the ones that avoid opportunistic behaviors, thus favoring the success of alliances.
Browning et al. (1995) [59]	The authors state that the success of alliances increases when the commitment between the people involved in the agreements goes beyond the contractual obligations. These ethical bonds create a moral community.

Table 1. Main contributions of alliances to corporate social responsibility.

In Table 1, we can see that different authors have focused their studies on the relationship between ethics and the success of strategic alliances between organizations. Such studies have been conducted using different approaches, such as corporate governance [60], stakeholders' theory [61], and the territorial perspective [62]. More recently, a big part of the literature on business cooperation has focused on the impact that social responsibility policies have on this type of agreements. The new studies emphasize aspects such as the formation, development, and control of the alliances [11,53,54].

To assess the impact that social responsibility measures have on cooperation agreements, Kliksberg [63] defined corporate social responsibility as a way of understanding the business world in which organizations are allies and drivers of environmental and social development. The author adds that capitalism and its individualistic ethics cannot be suitable to underpin social responsibility. Therefore, this system does not meet the needs of today's society. Ethics must be based on cooperation and mutual contribution [64].

Related to this aspect, Pulgar and Pelekais [11] identified the importance of strategic alliances as a tool to guarantee the effectiveness of social responsibility measures. By cooperating, partners in different or similar industries can unite their strengths in order to meet the needs and requirements of the society in which they operate. Additionally, Morales et al. [65] affirmed that social responsibility policies can be more effective if they are carried out through cooperation with entities that possess the key competencies for their implementation, that is, through strategic alliances with non-profit organizations. Consequently, these authors, together with Porter and Kramer [18], introduced the importance of alliances between for-profit and non-profit organizations for the creation of shared value.

The number of publications on strategic alliances between for-profit and non-profit organizations is still scarce and very recent. With regards to this subject, we can find papers that focus on diverse topics. Some of these topics are health and education [66], intersectoral alliances [67], cooperation agreements between private entities and non-profit organizations [68], intersectoral alliances that focus on corporate websites [69], alliances centered on the use of the balanced scorecard [70], cooperation agreements that focus on the implementation of marketing strategies [71], intersectoral alliances centered on reputation [72], alliances focused on strategic development [73], and alliances centered on the mediating role of altruism [74]. Table 2 summarizes the main contributions concerning the relationship between alliances between for-profit and non-profit organizations and corporate social responsibility.

Authors	Contributions
Muthuri et al. (2009) [75] and Porter and Kramer (2011) [18]	Hybrid corporations are those that are capable of creating shared value as the intersection between economic and social value.
	The authors state that these alliances are created as a response to the new interests of the stakeholders. In this sense, the work establishes three perspectives or approaches:
	(1) Transaction: occasional philanthropic actions that do not create value.
García et al. (2011) [14]	(2) Transition: continuous dialogue between the companies involved in the alliance but without considering the rest of the stakeholders.
	(3) Transformation: continuous dialogue between all of the stakeholders, thus creating shared value throughout the development of the alliance.

Table 2. Main contributions of alliances between for-profit and non-profit organizations to corporate social responsibility.

Authors	Contributions		
Berger et al. (2004) [66]	This paper studies the holding of events to raise funds for cancer research through an alliance between American Express and Share Our Strength (a non-profit organization against hunger) in 1991.		
Shumate and O'Connor (2010) [69]	This paper studies successful alliances between 11 private entities and 59 non-profit organizations based on a sample of 155 companies from the US Fortune 500 and 6965 NGOs.		
Hansen et al. (2010) [70]	The authors study the positive social impacts for local communities derived from the alliance between Merck, Ltd. Thailand (a subsidiary of a German pharmaceutical multinational) and local organizations.		
Arya and Salk (2006) [67], Rivera-Santos and Rufin (2010) [68], Liu and Ko (2011) [71], Im (2011) [72], Al-Tabbaa (2014) [73], and Rim et al. (2016) [74]	This work's contributions are related to the benefits obtained from alliances created to establish socially responsible practices.		

Table 2. Cont.

With regards to cooperation between for-profit and non-profit organizations, Porter and Kramer [18] stated that, on the one hand, for-profit corporations can embrace the social and environmental aspects predominating in the non-profit entities. On the other hand, the authors affirm that non-profit organizations can ensure their economic viability to continue operating. Then, the existing boundary between for-profit and non-profit organizations blurs. By cooperating, both entities complement their deficiencies, thus guarantying their survival. Therefore, these strategic alliances lead to the creation of shared value, which is based on the idea that organizations create economic and social value through their economic activity.

In the same line of thinking, Rodríguez et al. [16] focused on the impact that this type of cooperation has on both organizations. According to their studies, the positive impacts and benefits derived from strategic alliances between for-profit and non-profit organizations can be seen in three different factors: (1) for-profit organizations experience an increase in their number of partners as a consequence of improvement in their corporate image derived from the alliance; (2) employees from both entities undergo an increase in their motivation derived from a higher general and operative implication at work; and (3) these alliances promote the exchange of tangible and intangible resources and capabilities between both organizations. It is worth noting that this last aspect leads to a process of in-depth learning, which will remain with each one of the entities if the cooperation agreement ceases in the future. Moreover, the authors mention that this knowledge derived from the alliance will allow both organizations to compete in meeting the needs and requirements of every single stakeholder of the society in which the companies operate.

Concerning this topic, Carreras and Iglesias' [15] work shares some of the benefits explained above, but they added some new ones. On the one hand, with regards to the benefits experienced by capitalist organizations, they included (1) an improvement in their internal processes and the potential for innovation in their human resources policies, (2) the retention of employees due to an increase in their motivation, (3) higher visibility of the organization's value, (4) increase in the company's awareness of the reality of the society in which they operate, and (5) the implementation of social responsibility measures becoming easier thanks to the knowledge obtained from the non-profit entity. On the other hand, the positive impacts experienced by the non-profit entities are (1) access to economic and technical resources, (2) creation of new products and services as a consequence of the know-how acquired, (3) increase in their channels of distribution, (4) access to a broader number of partners due to the networks of the for-profit company, and (5) increase in their visibility.

Therefore, we can conclude that different benefits exist for both types of entities, which explain the increase in the importance of this type of alliances during recent years [76]. Nevertheless, the importance

of these agreements does not reside in this aspect, since alliances between for-profit organizations also present benefits for the entities involved. Therefore, many papers focus on the impact that these agreements constitute for the community where they are developed [13]. Concerning this aspect, Abenoza et al. [17] indicated that there are three positive impacts for the community derived from this type of cooperation.

In the first place, they stated that entities can solve a social problem more efficiently due to the exchange of resources and capabilities that takes place between them.

In the second place, the authors mention that organizations are able to generate social innovation as well [17]. As a consequence of the cooperation between for-profit and non-profit entities, organizations can develop new products, services, and technologies with significant social impact. That is, these alliances lead to the creation of innovative offers aimed at solving the social problems mentioned above. On the one hand, the non-lucrative entity detects the environmental or social challenge by bringing its knowledge about this aspect, while on the other hand, the for-profit organization brings its professional experience related to product or service development.

Finally, the third community benefit that Abenoza et al. [17] indicated in relation to these types of alliances is related to the local and global changes they promote. The authors referred to these cooperation agreements as strategic alliances developed to promote ethical practices. They named these practices "global action networks", which are described as coalitions created to promote not only local changes but also global ones. These agreements are based on (1) shared learning and (2) the establishment of innovative networks emerging from the technological exchange between the organizations involved. According to the authors, these agreements are created to encourage social transformation. Such social transformation is the reason why these alliances are not restricted to the local community in which the organizations operate, but they pretend to have a positive integral impact through innovation.

In this line of thinking, Callejas and Salazar [77] indicated that the exchange of knowledge and technology that takes place between for-profit and non-profit organizations leaves a communitarian footprint globally and locally. Furthermore, the authors added that this footprint can be seen at the social and environmental levels. These types of alliances allow, through innovation, the development and implementation of new models of production and distribution of products and services designed for the common good. Moreover, these new models are environmentally friendly, and they do not need to be limited to the locality in which the organizations are established.

As a conclusion, we can state that the transcendence of the strategic alliances between for-profit and non-profit organizations resides on the impact that they have on the community. The creation of economic and social sustainable value is only possible through this model of cooperation since these strategic alliances allow social development through the economic one and vice versa [78].

2.2. The Economy for the Common Good Organizational Model

The ECG model was created in the year 2010 during the financial crisis that began in Central Europe in 2008. The model was created by Christian Felber, professor of Economics at the University of Vienna and social activist of Attac. Hence, the model has social origins, being created as an alternative perspective to the current global economic model. The underlying topic that this model suggests is that economic growth and money cannot be considered an end or an objective; these must be instruments used in order to create social welfare and to improve the quality of people's lives. From this perspective, Felber proposes the substitution of "for-profit" by "common good" and the replacement of competition by cooperation [26]. The substitution of "for-profit" by "common good" does not imply that organizations are not going to obtain benefits; this means that entities must favor the general interest through the creation of social value.

Cooperation and competition are two factors that have been analyzed together in business literature. Firstly, they were considered two opposing terms (when cooperation increases, competition decreases) [31–33]. Later, they were contemplated as two concepts directly related (cooperation favors

competition) [4]. Nevertheless, it is important to mention that competition and cooperation are not incompatible concepts; they can be supplemented by what is known as co-petition [79]. This concept arises from game theory application. According to this theory, companies must follow a win-win strategy in which success in the market does not require the failure of competing companies, as there may be multiple winners [80–84]. Cooperation is used by organizations to be more competitive without the need to reduce or eliminate competence.

The ECG model was created as a proposal for a global economic model. Nevertheless, its implementation focuses on the organizational sphere. With this purpose, the model uses a tool known as the Common Good Balance Sheet (CGBS), in which its utility is to assess the impact that an organization produces on society (social impact) and the planet (environmental impact) [85]. The CGBS works as a complementary tool to the balance sheet and to the profit and loss statement, which measure the economic and financial structure of an entity. The CGBS consists of the Common Good Matrix, and the company's improvement plan.

The Common Good Matrix is a strategic matrix that relates the four fundamental values of the ECG model (human dignity, solidarity and social justice, environmental sustainability, and transparency and codetermination) with five main groups of stakeholders (suppliers, owners and financial service providers, employees, customers, and their social environment). Through the matrix, organizations can quantitatively measure their contribution to the common good. According to the score range obtained in the matrix, the company can be placed at the following rating levels: beginner (0–1), advanced (2–3), experienced (4–6), and exemplar (7–10). The highest possible score is a thousand points. Furthermore, this matrix can be applied to any organization, whether they are private (for-profit and non-profit corporations) or public. The matrix structure can be seen in Table 3.

Value/ Stakeholders	Human Dignity	Solidarity and Social Justice	Environmental Sustainability	Transparency and Codetermination
A Suppliers	A1. Human dignity in the supply chain	A2. Solidarity and social justice in the supply chain	A3. Environmental sustainability in the supply chain	A4. Transparency and codetermination in the supply chain
B Owners, equity, and financial service providers	B1. Ethical position in relation to financial resources	B2. Social position in relation to financial resources	B3. Use of funds in relation to social and environmental impacts	B4. Ownership and codetermination
C Employees, including coworking employers	C1. Human dignity in the workplace and working environment	C2. Self-determined working arrangements	C3. Environmentally friendly behavior of staff	C4. Codetermination and transparency within the organization
D Customers and other companies	D1. Ethical customer relations	D2. Cooperation and solidarity with other companies	D3. Impact on the environment of the use and disposal of products/services	D4. Customer participation and product transparency
E Social environment	E1. Purpose of products/services and their effects on society	E2. Contribution to the community	E3. Reduction of environmental impact	E4. Social codetermination and transparency

Table 3. The Common Good Matrix.

Source: https://www.ecogood.org/apply-ecg/common-good-matrix/. Note: This matrix corresponds to the 5.0 version, created in 2017. The original matrix (from 2010) has been modified based on the changes that have been introduced over time. As a consequence of these changes, nowadays, there are five different versions of the matrix. The first one was implemented through 100 German and Austrian organizations in 2010.

2.3. Cooperation in the ECG Model

Cooperation appears implicitly and transversally in every single one of the 20 topics comprised within the matrix. In order to quantify these topics, the matrix uses different types of indicators. Detailed information on the matrix structure and the process through which the scores are obtained

can be found on the website of the International Association for the Common Good Economy [86]. By analyzing each one of its indicators, we can identify those aspects related to cooperation which appear explicitly in the matrix. Therefore, throughout the study of the ECG Matrix, we can find references to cooperation in both ways, implicitly and explicitly. The next section Section 2.3. of the work focuses on these references.

In the first place, by studying the first interest group of the matrix (suppliers), we analyze the relationship between the organization and the suppliers across the supply chain. In this case, alliances have significant relevance, since we consider these types of relationships as vertical alliances within the supply chain. This block of the matrix introduces the concept of "sustainable supply chain". A sustainable supply chain is one in which a situation of abuse of market power between the company and its suppliers does not take place. Moreover, it includes the establishment of cooperation agreements with local suppliers, thus favoring proximity sourcing and relationships.

Furthermore, this block includes the conditions of the cooperation agreement between the company and the supplier. Sustainability criteria bring these conditions, which are specified through the four values placed horizontally in the matrix. Such values represent a sustainable supply chain management. Cooperation is reflected explicitly in the fourth value corresponding to "A.4. Transparency and codetermination in the supply chain". In this case, the ECG model suggests establishing cooperative relations with suppliers based on transparency and participation.

In the second place, by studying the second interest group of the matrix (owners and financial service providers), the company's relationship with its financial providers (primarily credit institutions) is analyzed. In this block of the matrix, horizontal alliances between the organizations and the financial entities take place. These alliances—when we apply the sustainability criteria—can be understood as coalitions with ethical and social banks. This aspect implies the establishment of cooperation agreements with financial providers, which are based on socially responsible investments. In this case, the company can participate in solidarity-based financing in order to implement projects with a positive social and environmental impact. The implementation of these projects will be conducted employing subordinated loans, microcredit, and crowdfunding or by providing financial aid directly. It is important to mention that this matrix includes a negative indicator (it implies subtracting points from the matrix): the possibility that the organization participates in a hostile takeover bid.

In the third place, by studying the third interest group of the matrix (employees), the relationships between people by using tools based on participation and co-decision are analyzed. In this case, cooperation is studied from an internal perspective known as intra-cooperation. This one uses different types of methods, such as cooperative work or collaborative decision-making. This block of the matrix shows that an organizational culture based on respect, trust, and appreciation fosters cooperation between people, thus including an open, transparent, and bottom-up communication system.

Moreover, the matrix includes indicators related to diversity—age, gender, ethnicity, and religion—and equal opportunities. Nevertheless, the aspect that focuses the most on cooperation is the one related to transparency and codetermination or internal democratic participation. In this sense, the first aspect valued is transparency or access to the company's relevant information; by accessing information, people can build their own opinion and can participate actively. A second important aspect to consider is the legitimacy of the organization's management; employees are the ones who must legitimate directors through dialogue, contribution, and participation. There is also a third aspect related to participation of employees in the decision-making process of the company; workers can give ideas, and they can take action in the entity's decision-making process. To this end, ECG theory suggests the following concrete actions: making bottom-up decisions, sharing responsibility, covering all levels of the organization, and applying direct democracy in relevant decisions.

In the fourth place, by studying the fourth interest group of the matrix (customers and other companies), the relationship of the company with its clients (favoring their participation in those decisions that directly affect them) and with other entities (enhancing inter-cooperation) is analyzed.

With respect to the customers, ECG theory proposes to maintain a transparent, honest, and horizontal relationship with them. Furthermore, the model suggests facilitating their access to the purchasing process by actively participating in decisions that affect them. According to ECG theory, such access could be promoted by creating customer groups, working committees, etc. Clients can participate in the improvement or development of existing products or services as well as in facilitating their diffusion among customers.

With regards to cooperation with other entities, collaboration should be based on an attitude of appreciation and equal and horizontal treatment. This section of the matrix is based on three aspects: (1) considering organizations in the same sector as complementary in the market, (2) working with other companies to offer joint solutions to the customers, and (3) offering disinterested support to other entities in a situation of need. These three aspects form a condition that, in the theoretical framework, was referred to as co-petition. Some of the matrix indicators related to co-petition are (1) the percentage of time and resources spent on collaborative product and service development, (2) the percentage of time aimed at enhancing cooperation with other organizations from the same or different market segments and sectors, and (3) the number of civil society projects shared with other entities. Finally, solidarity towards other organizations is implemented by promoting mutual disinterested assistance through the exchange of employees, assignments, financial resources, and technology. In this block of the matrix, hostile behavior towards other organizations, the abuse of its competitive position in the market, and seeking win-lose strategies are activities negatively assessed.

In the fifth and last position, by studying the fifth interest group of the matrix (the social environment), the relationship of the company with society is analyzed. This block focuses on the relationship between the organization and the region where it develops its economic activity. Here, we can include alliances between for-profit organizations and NGOs or other non-profit entities as well as cooperation agreements with other groups of interest such as neighborhoods, ecologist associations, etc. The theoretical aspects previously analyzed in this work with regards to alliances between for-profit and non-profit organizations can be applied to this block of the Common Good Matrix.

Moreover, this section of the model comprises different ways to collaborate with the community by means of voluntary contributions. Such contributions can be monetary or in-kind. They can also take place by using the company's network to support civil society projects. Besides, cooperation with interest groups of the community, such as local institutions and NGOs, represent another way to collaborate.

These contributions and impacts are based on transparency; they are possible because the organization provides to the external interest groups truthful and complete information with regards to the entities' actions and their direct effects on the social environment. In connection with this, direct and active participation with the community's interest groups should be encouraged by consulting them and by involving them in the decision-making process of the company, especially in those aspects that directly affect the social environment. Information manipulation and lack of transparency are valued negatively in the matrix.

By studying the indicators proposed and reflected in the Common Good Matrix, we can quantify the role that cooperation plays in the implementation of the ECG model.

3. Method

The present study adopts a qualitative methodological approach, more specifically, the case study method. According to Yin [87], this methodology is the most appropriate one due to the critical and peculiar character of the cases and the unique object of study that they comprise. This approach enables investigating and drawing conclusions in a way that would not be possible otherwise. Moreover, multiple case studies (research in which several cases are analyzed at the same time) enables the investigation and comparison of examples and facts so that, finally, reality can be described [87]. In the same line of thinking, authors such as Rule and John [88] state that it is possible to test and affirm theories with the case study method. Their study indicates that qualitative methodologies can test

and verify theories through the study of cases employing deductions that move from the general to the specific.

This generalization is described by Yin [89] as "analytic generalization", a process that takes place when a theory can be used as the basis to compare the empiric results obtained with the case study.

Since the final objective of this work is to demonstrate the theories exposed in the theoretical framework, this work conducts an empirical study consisting of the analysis and comparison of two different cases. These cases have been selected for two main reasons: (1) visibility of the participating organizations and (2) the amount of information published with regards to the cooperation agreements between them.

The study of the two selected cases has been conducted based on the information obtained in such cases. We have chosen the cases based on the impact that they have on the community and the ease of getting sufficient and adequate information. This information has been obtained from two different types of sources: (1) from the institutional websites of the entities that are part of the alliances and (2) from different papers published on academic journals and other public documents.

In order to meet the work's purpose, the study is divided into three sections: (1) an individual study of the strategic alliance between Grupo Vips and Fundación Hazlo Posible, (2) an individual study of the strategic alliance between Danone Foods and Grameen Bank, and (3) a comparative analysis of both cases with the aim of discerning if there is a common pattern concerning the implications of these cooperation agreements for the community and their relationship with the ECG.

4. Case Study Participants Description

4.1. Grupo VIPS–Fundación Hazlo Posible Case Study

On the one hand, Grupo Vips is a family business founded in Madrid in 1969, when Plácido Arango (the founder) decided to open the first establishment under the name Vips. Although initially the organization had only one chain of restaurants, nowadays, the group is formed by six big chains and ten restaurants with more than 350 establishments and a workforce of more than 10,000 employees [90].

On the other hand, Fundación Hazlo Posible is a private non-profit organization founded by José Martín and Catalina Parra in 1999 to stimulate the involvement of society in solidarity and social causes by using information and communication technologies. With this objective, they created an online portal web under the name Canalsolidario.org [90]. This portal web was intended to work as a bridge between the activities of the NGO and the society. Today, Canalsolidario.org is a digital newspaper specialized in social information, while the foundation is focused on three action lines: awareness-raising and communication, volunteering, and knowledge management. The entity has a workforce of 17 employees and a census of 500 volunteers who support the organization guided by the values of commitment, diversity, energy, and creativity [90].

Cooperation between both organizations began when Maite Arango (vice-president of the Vips Group Council) and Elena Acín (director of *Fundación Hazlo Posible*) met at the first Internet Fair in Madrid in 1999. Acín attended the fair to present their new portal web *Canalsolidario.org* with the aim of accessing private funding, while Arango was looking for a social project to which *Grupo Vips* could commit [90]. After several meetings, both entities decided to start to collaborate on the portal web *Canalsolidario.org* in 2000. When cooperation started, the portal was renamed *Hacesfalta.org* after *Grupo Vips* made an initial investment of $106.000 \in$. The decision to start cooperating was taken quickly since both organizations shared the same goal. Such a goal was to increase the visibility of a new type of volunteering in Spain, that is, virtual volunteering. With this purpose, both entities unified their knowledge: *Grupo Vips* contributed with its customer service experience and *Fundación Hazlo Posible* contributed with its consulting skills. After this first contact, during the following years, in order to avoid complications related to compliance with the agreements, both entities set up formal governance mechanisms such as the signing of annual agreements and the submission of monthly reports to monitor the development of the project [90].

In 2004, *Fundación Hazlo Posible* launched a new corporate volunteering program as an extension of *Hacesfalta.org* in which different organizations were able to participate. To manage this program, *Grupo Vips* created a new corporate social responsibility department so that the relationships between the organizations become formal and informal. Nevertheless, it is important to mention that, in addition to these formal administrative procedures, two main aspects explain the long-term nature of this alliance. These aspects are (1) a strong level of engagement and interactions between the entities, and (2) trust, which has been an essential governance mechanism for the development of this cooperation agreement.

One year later, in 2005, both entities developed a project known as *Brazos Abiertos* (open arms). This new project, which was part of the corporate volunteering program, was aimed at incorporating new employees in their organizations and at integrating immigrants arriving in Spain into the Spanish society. Due to the growth of the different projects carried out, organizational changes were necessary. These adjustments caused a renewal in the management positions of both organizations, and they generated the incorporation of several new skilled employees into the workforce [90].

Over the next three years, as a consequence of the financial crisis of 2008 and due to the rapid growth of the alliance, *Fundación Hazlo Posible* detected the need to look for new sources of financing. With this purpose, in 2010, both organizations signed a three-year contract consisting of two aspects: (1) *Grupo Vips* would reduce the amount of money they invest in the non-profit organization, and (2) *Fundación Hazlo Possible* would look for other sponsorships and advertising contracts. Thus, in 2010, the strategic alliance was strengthened by opening the non-profit organization to new sources of funding [90].

Nowadays, both entities work together on the *Hacesfalta.org* project and on a new one incorporated in 2019 as part of the corporate volunteering program, named *Red Talento Que Impacta* [91]. This new project promotes strategic relationships between for-profit and non-profit organizations to foster the transfer of knowledge and learning between both types of companies. Therefore, *Red Talento Que Impacta Que Impacta* enhances and strengths the creation of social value in the community [92].

Concerning the impact of this alliance on the community where the organizations are established, we can find different benefits. First of all, since the creation of the first volunteering portal web, both entities have contributed to enhancing the volunteering and social responsibility culture in Spain. Since the creation of this cooperation agreement, hundreds of non-profit and for-profit organizations have actively participated in several solidarity initiatives, thus fostering the solidarity and corporate responsibility values [90]. Furthermore, this new organizational culture fosters communication between the different departments of *Grupo Vips*, thus leading to an increase in the motivation of employees, alongside a more robust leadership of its professionals [90].

In the second place, the strategic alliance has also contributed to technological development. As a result of the cooperation agreement, the portal web *Canaldolidario.org* and other following projects were created and intensively developed, initiatives that had a strong social repercussion. Therefore, we can state that, through this type of cooperation agreement, entities can develop new innovative products and services aimed at solving social problems [17]. Moreover, this technological development had also a positive impact on the organizations involved in the cooperation agreement [90]. This pioneering technological development has contributed to strengthening the reputation and corporate image of *Grupo Vips*. As stated by Rodríguez et al. [16], an improvement in the corporate image derived from the alliance enables the recruitment of new partners, alongside an increase in the reputation of the company. Therefore, we can conclude that both entities have benefited to a great extent since the cooperation agreement was established.

On the one hand, as a result of an increase in its visibility and a significant rise in its economic resources, *Fundación Hazlo Posible* has experienced a considerable growth of its partners: nowadays, the non-profit entity has 53 allies, and it is part of ten social networks [93]. On the other hand, *Grupo Vips* has undergone a clear improvement in its corporate image. Currently, the group is considered one of the most well-known businesses within its sector as a consequence of its social activity [94]. Since the establishment of the alliance, the for-profit organization has been awarded

several distinctions provided by institutions such as MERCO (Spanish Corporate Reputation Monitor) or the organization *Fundación Empresa* [94]. Moreover, *Grupo Vips* is invited to academic forums to talk about the key to its success. Besides, the entity is cited in different publications and academic journals related to this subject. These recognitions have fostered the company's growth: before the alliance, the group had a workforce of 2800 employees, while nowadays, it has 9300 workers distributed among the 400 establishments and six chains that the business currently has [94].

Therefore, we can observe that, through cooperation between for-profit entities and non-profit organizations, companies bring benefits to society and the cooperating businesses as well. Thanks to the collation, *Fundación Hazlo Posible* and *Grupo Vips* constitute a culture based on diversity, innovation, teamwork, and efficiency, a culture that benefits every single group of interest of the agreement [90]. Through the strategic alliance, both organizations share resources and capabilities, thus allowing the for-profit organization access to those it did not possess before the alliance. Such resources enable the transfer of knowledge that makes the development of the portal web possible, which brings benefits for both entities and the community.

Throughout their repeated joint activities based on trust, both organizations created a unique synergy that brought them a sustainable competitive advantage, which allowed the alliance activity and its benefits to last over time. Porter and Kramer [18] stated that this sustainable competitive advantage is based on the shared value creation: through cooperation between for-profit and non-profit organizations, entities create social value benefiting the economic activity of the partners involved and extending the positive impact they have on society. Thereafter, we can observe that cooperation between *Grupo Vips* and *Fundación Hazlo Posible* assists in the creation of social value and the common good, thus fostering implementation of the ECG model.

Table 4 summarizes the main contributions of the case study on implementation of the ECG model from a cooperation perspective.

Value/ Stakeholders	Human Dignity	Solidarity and Social Justice	Environmental Sustainability	Transparency and Codetermination
A Suppliers				
B Owners, equity and financial service providers		Financing of social projects	Financing of environmental projects	
C Employees, including coworking employers	Diversity in employment	Voluntary work	Lower impact due to the ICT use	Employee motivation and participatory culture
D Customers and other companies				
E Social environment	Culture of volunteerism	Community aid and support	Lower impact due to the ICT use	Environmental participation

Table 4. Contributions of the strategic alliance *Grupo VIPS–Fundación Hazlo Posible* to the Economy for the Common Good (ECG) model.

4.2. Danone Foods–Grameen Bank Case Study

On one side, *Danone Foods* is a food business founded by Isaac Carasso in 1919 to improve people's health through progress in the process of fermentation. Although the first factory of the group was established in Barcelona in 1929, the company moved to France, where they inaugurated the first factory for the production of yogurts. Nowadays, the corporation has four different business lines: dairy products, water, infant nutrition, and medical nutrition. Moreover, the group operates in more than 130 countries spread over the world [95].

On the other hand, *Grameen Bank* is an entity created by Muhammad Yunus in Bangladesh in 1940 to offer microcredits to the most impoverished population of the region without the need for guarantees. Yunus idea was to provide funding capacities to the poorest people with terms and conditions they could afford. With this action, Yunus intended to carry out small activities that could contribute to the development of the community. Currently, *Grameen Bank* has 2568 branch offices with a workforce made up of 21,751 employees. Moreover, the organization has a base of nine million borrowers in more than 80,000 villages. Besides, the methods employed by *Grameen* are applied to projects in more than 58 countries such as The United States, Canada, France, Norway, and The Netherlands, among others [96].

The strategic alliance between the two organizations dates back to 2005 when, while visiting the city of Paris, Yunus received an invitation from Franck Riboud (CEO of *Danone Group*) to meet. During the meeting, the founder of *Grameen* suggested to Riboud the creation of a cooperation agreement between the two entities to jointly address the child malnutrition in Bangladesh, particularly in rural areas of the country. To this end, in 2006, the joint venture *Grameen Danone Foods* was created as a social enterprise [97]. This business model focused on the resolution of social and environmental issues. For that reason, the benefits obtained from the social enterprise activities were reinvested in the organization, so that a feedback loop that generates growth while contributing to development was created. Even though this social enterprise must generate profit that can be reinvested into the organization to assist in a continuous process of progress and progress [98].

Hence, the social joint venture *Grameen Danone Foods* started to produce a new yogurt under the name of *Shoktidoi* ("strong yogurt"). The enterprise manufactured this product with essential nutrients, as it is specifically designed to fill the nutritional deficiencies of children in Bangladesh. The product had a selling price of eight cents, being affordable for the most impoverished population since it is intended for children in need [99].

In 2006, they established the first production plant in the city of Bogra, a factory 50 times smaller than the standard plants of *Danone Group*, which supplied a rural population spread over a 30 km perimeter around the *Shoktidoi* factory [100]. Since the establishment of this plant, the joint venture has invested around 700 thousand dollars in the region. Besides, more than seven million people have benefited from access to health, hygiene, nutrition, and employment [101].

With regards to the impact that the strategic alliance *Grameen-Danone Foods* has brought to different groups of the community in which the enterprise is located, it is worth noting that this cooperation agreement shows benefits in the environmental, economic, and social spheres.

Firstly, to reduce the environmental footprint of the social enterprise, the plant located in the city of Bogra works with solar energy employed to heat the water and to clean the facilities [97]. This aspect proves that, by using this renewable energy source, the social entity contributes to putting into practice the ECG model. In this case, by reducing the environmental effects of its activity, they focused on the ecological solidarity and sustainability aspects [27].

Secondly, concerning the economic impact of the strategic alliance in Bangladesh, it is important to mention that the social enterprise has become a significant source of income creation for the Bogra region and its surrounding areas. On the one hand, 475 farmers located in the area around the plant provide the milk used for the production of the yogurts. Moreover, these yogurts are manufactured by 30 local employees, and they are distributed by 250 micro-entrepreneurial women from the area who sell the products door-to-door. In total, *Grameen Danone Foods* employs 1600 employees from the area in which the plant is located [97].

Hence, we can observe that different agents of the community are involved in the manufacturing and distribution processes of the nutritious yogurts. Consequently, the joint venture improves the nutrition of children at the same time that it creates a business fabric that offers opportunities for employment and, therefore, for progress to different sectors of the community. As with the environmental sphere, through its activity, *Grameen Danone Foods* also contributes to implementing the EBC model in the economic sphere. In this case, the joint venture focuses on social justice and human dignity values. Such values are observed in the extensive contribution to the community and social impact that the social enterprise represents [27].

Thirdly, concerning the social impact of the cooperation agreement in the community, we can find the following benefits. On the one hand, the plant situated in Bogra produces 100,000 units of yogurt daily, which means that 100,000 children from the area receive nourishment that covers 30% of their daily needs for vitamin A, iron, iodine, and zinc [99]. Thus, the strategic alliance achieves one of their main objectives: the improvement of nutritional conditions of children in Bangladesh, especially those in rural areas.

On the other hand, the coalition has helped to eradicate the exclusion of the poorest groups of the community by creating a large number of local jobs, as was mentioned above. This job creation has favored access of people from rural areas to goods and services they could not afford previously, such as access to finance, drinking water systems, or electricity. These aspects have led to a significant reduction in inequality of the rural population with respect to more developed areas. Therefore, the joint venture also contributes to the ECG model with the values of justice, human dignity, and solidarity [85], thus actively promoting development of the community.

Thereafter, thanks to the strategic alliance, *Grameen Bank* and *Danone Foods* created shared value, value based on the idea that, through its economic activity (in this case, the manufacture of yogurts), the organizations involved in the agreement create both economic and social value [18]. Furthermore, with the creation of a social enterprise, both firms diluted the historical definition of each type of entity. They created a hybrid organization in which the for-profit company covers the social and environmental aspects predominating in the non-profit organization, and this, in turn, ensures the economic viability necessary to continue operating [18].

Finally, it is worth noting that both organizations have also benefited from the strategic alliance. On the one hand, since the social enterprise was established, *Danone* has experienced an increase in its number of partners. Moreover, nowadays, the for-profit entity can implement social responsibility policies quickly as a consequence of the transfer of knowledge derived from the alliance, thus strengthening its corporate image [95]. The creation of the social enterprise *Grameen Danone Foods* became the foundational project of *Danone Communities*, a social business network aimed at assisting local entrepreneurs with funding and formation [99]. Currently, *Danone Communities* consists of 11 social businesses which include several for-profit organizations [99]. This network has significantly improved the group's brand positioning. According to the *Reptrak index* (an indicator of corporate reputation issued by the Reputation Institute) [102], during the period 2010–2014, *Danone Group* was positioned first in the ranking. The corporation was recognized in 2015 as the company with the best reputation in the food and beverage sector [100].

On the other hand, since the establishment of the strategic alliance in 2006, *Grameen Bank* has founded seven social enterprises [103], thus accessing a vast number of people in need.

Therefore, we can conclude that, through this cooperation model in which organizations shared their different resources and capabilities, the social enterprises derived from the alliances can solve a social issue more efficiently than operating separately [17]. The social enterprise *Grameen Danone Foods*, by producing and locally distributing affordable yogurts to the population, enabled an improvement in the nourishment of the children. Moreover, it generated an ecosystem in which different sectors of the community participated. This factor increases the economic capacity of the rural people from the area so that it generates social and economic development there. Thus, we can observe that the joint venture *Grameen Danone Foods* contributes to the implementation of the ECG model through its cooperative activity in which capital is not the goal but an instrument for social transformation [26].

Table 5 summarizes the main contributions of the case study to implementation of the ECG model from a cooperation perspective.

Value/ Stakeholders	Human Dignity	Solidarity and Social Justice	Environmental Sustainability	Transparency and Codetermination
A Suppliers	Support to local suppliers	Fair prices to local suppliers	Working with sustainable suppliers	Collaboration with local suppliers
B Owners, equity and financial service providers	Financing of social projects	Ethical banking promotion		Collaboration with ethical finances criteria
C Employees, including coworking employers	Equality and decent employment	Combating social and labor exclusion		Local employment creation
D Customers and other companies	Products at affordable prices	Local entrepreneurship support		Collaboration with the community
E Social environment	Child nutrition	Poverty eradication	Solar energy use	

5. Results

After analyzing the cooperation agreements, we can state that there is a common pattern between both cases concerning the benefits provided to the community. The strategic alliances studied solve a social issue by sharing their different resources and capabilities.

On the one hand, *Grupo Vips* (by providing funding and contacts) and *Fundación Hazlo Posible* (by contributing with its social responsibility knowledge) expand a portal web aimed at the insertion of organizations and different sectors of the society in volunteering aspects. Hence, we can observe that *Grupo Vips–Fundación Hazlo Posible* solve a community problem through social innovation [17]. In this case, the organizations involved contribute to technological development through their portal web, a technology that has a significant impact and social benefit.

On the other hand, *Danone Foods* (by contributing with its food processing knowledge) and *Grameen Bank* (by providing its social enterprise and Bangladesh market knowledge) improve the nourishment of the children in the area. Moreover, the joint venture assists in the economic progress of Bogra by creating employment.

Furthermore, the entities involved in the cooperation agreements experience benefits for themselves. *Grupo Vips, Fundación Hazlo Posible, Grameen Bank,* and *Danone Foods* have benefited from a significant increase in their number of partners. Moreover, their networks have been expanded as a consequence of their greater visibility in the community. Likewise, since the establishment of the alliance, *Grupo Vips* and *Danone Foods* are considered some of the most reputable companies within their sectors [94,95].

Finally, it is worth noting that these cooperation agreements lead to the "Creating Shared Value" approach referred to by Porter and Kramer [18]. Both strategic alliances—*Grupo Vips–Fundación Hazlo Possible* through social innovation and *Grameen Bank–Danone Foods* through economic progress—generate social benefits which allow the creation of economic benefits. As a result of their work together, the organizations involved in the strategic alliances can redefine the for-profit and non-profit entity concepts. Thus, the cooperating enterprises become a hybrid organization that ensures its economic viability and survival by advancing business competitiveness while fostering their economic and social environment [18].

Table 6 summarizes the similarities between the two case studies analyzed. As we can observe, both cases have a positive impact on the social environment. Such impacts can be seen in three

different aspects: (1) the interaction of the organizations with the social agents, (2) the reduction of their environmental footprint, and (2) their support to the community. Furthermore, by promoting decent working conditions, socio-occupational integration, and the culture of participation, the entities have a positive impact on the employees as well. Finally, both agreements favor the funding of social projects, thus contributing to the values of solidarity and social justice.

Value/ Stakeholders	Human Dignity	Solidarity and Social Justice	Environmental Sustainability	Transparency and Codetermination
A Suppliers				
B Owners, equity and financial service providers		Financing of social projects		
C Employees, including coworking employers	Decent employment	Socio-occupational integration		Culture of participation
D Customers and other companies				
E Social environment		Community support	Reduced environmental impact	Interaction with the environment

Table 6. Similarities between the two cases according to the ECG Matrix.

Therefore, we can observe that both cooperation agreements contribute to implementation of the ECG model. On the one hand, both cases place cooperation ahead of competition (basic principle of the ECG model) [27]. On the other hand, the strategic alliances operate meeting the values considered as contributors to the common good: human dignity, solidarity, sustainability, social justice, and participation [26]. *Grupo Vips–Fundación Hazlo Posible* and *Grameen Bank–Danone Foods* generate social value and shared value, thus contributing to implementation of the ECG model.

6. Discussion

From the present study, we can conclude that cooperation is a crucial instrument when implementing sustainability-based management models. Such sustainable models require the dominance of its three dimensions (economic, social, and environmental), dimensions that not all companies possess.

To this end, strategic alliances enable organizations to access the resources and capabilities needed to develop the three dimensions of sustainability that they cannot attain individually. From this perspective, alliances between private for-profit organizations and non-profit entities become a particularly suitable tool for the implementation of corporate sustainability. Thus, for-profit organizations possess the resources needed to ensure economic sustainability while NGOs provide adequate resources and capabilities for social and ecological sustainability. Hence, as stated by Porter and Kramer [18] within their "Creating Shared Value" approach, both types of entities can create shared value in which economic, social, and environmental value reinforce each other. Furthermore, from the authors' approach, we can conclude that those entities with a higher probability of survival are the ones referred to as hybrid organizations since such organizations are capable of creating economic and social value cooperatively.

As cited in the theoretical framework, during the last years, papers related to this approach to cooperation are being published. While there are not many publications that study this aspect,

17 of 22

there are already some papers that analyze this perspective with interesting results and conclusions. In this regard, this work is a clear contribution to this line of investigation. On the one hand, this work analyzes strategic alliances between for-profit and non-profit organizations, thus reinforcing the literature in this field. On the other hand, the study connects alliances with corporate sustainability models, thus providing a clear innovation in this field of study. In particular, these contributions are driven by studying the impact that cooperation has when implementing the Economy for the Common Good model [27,85,104].

From the empirical study conducted throughout this work, we can state that there is a direct and positive relationship between cooperation and the implementation of the Economy for the Common Good model. By analyzing the case studies, we can affirm that strategic alliances favor the creation of social and environmental value for the different stakeholders of the organization. By using the Common Good Matrix, we have identified the actions of the alliance that create social value. We observe that the positive impacts derived from the cooperation agreements concentrate on some stakeholders such as the social environment, the employees, and the funders. In both case studies, organizations create social value by means of financing social projects, generating a participative corporate culture, and supporting the community. Furthermore, the matrix includes aspects such as the eradication of poverty and the socio-occupational integration of some groups, which are essential priorities of these types of alliances.

With these elements, several values of the ECG model are promoted. Such values are human dignity, solidarity and social justice, environmental sustainability, and transparency and codetermination. Likewise, one of the cooperation agreements studied also creates social value for suppliers, customers, and other companies. Moreover, it is important to highlight how alliances foster local suppliers and social entrepreneurship in communities where the cooperation agreements are developed.

It is worth noting that the literature on strategic alliances between for-profit and non-profit organizations is not developed in-depth yet. On the one hand, according to the classification carried out by García et al. [14] regarding the three different approaches that this type of alliances can adopt, we can conclude that most of these cooperation agreements are still in the first stages. Thus, the majority of these alliances focus on the transition approach, which is based on occasional philanthropic actions. Therefore, such alliances neither create shared value nor common good value.

On the other hand, the most progressive alliances are based on a transitional approach in which continuous dialogue between the for-profit and the non-profit organization takes place. Nevertheless, this perspective does not consider the rest of the stakeholders, thus creating shared value but not contributing to the creation of value for the common good. To reach the common good, alliances must include the transformation approach mentioned by García et al. [14], since this one is based on direct and active participation between all of the stakeholders that are affected by the alliance. From this point of view, the ECG model can help to strengthen this type of cooperation agreement. Consequently, not only the ECG model can benefit from the alliance but also, at the same time, these types of alliances can be enhanced through the ECG model. Therefore, there is a direct and reciprocal relationship between both.

7. Conclusions

The first purpose of this work was to establish a theoretical relationship between strategic alliances between for-profit and non-profit organizations and the ECG model. In this sense, the present work contributes to the literature on both topics. On the one hand, this work provides a new perspective on the study of this type of alliances. Firstly, we analyze the motives why organizations establish this type of cooperation agreement while connecting them with ethics and corporate social responsibility. Secondly, we analyze the impacts and benefits that these alliances create for the cooperating entities and for the community in which they operate.

On the other hand, this work provides a new perspective to the study of the ECG model, since this paper analyzes cooperation as one of the primary keys of the model. Although this model suggests

interfirm cooperation as a determining factor for business success, there is no literature on this subject. Therefore, this work is the first theoretical study that provides this ECG model approach.

The second objective of this work was to establish a practical application of the theoretical relationship between strategic alliances between for-profit and non-profit organizations and the ECG model. To this end, an empirical study based on the analysis of two case studies has been conducted. In this sense, our contribution is innovative and original, as there are no published empirical studies on this subject.

Finally, although the present work provides multiple case studies, it may be of interest to develop future researches concerning alliances between for-profit and non-profit organizations so that the taxonomy described can be expanded. Besides, it might be relevant to carry out an empirical study based on a quantitative analysis in which relationships between the different types of alliances and their social and environmental impacts can be analyzed.

A new paper on this topic could consist of application of the Common Good Matrix to a specific alliance between for-profit and non-profit organizations. Thus, the matrix indicators will be applied to quantitatively measure the contributions of these alliances to the common good.

Author Contributions: This article is a joint work of the two authors. C.T. and J.R.S. contributed to the research framework, literature review, empirical analysis, and conclusions. All authors read and approved the final manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Buckley, P.J.; Glaister, K.W.; Husan, R. International Joint Ventures: Partnering Skills and Cross-Cultural Issues. Long Range Plan. 2002, 35, 113–135. [CrossRef]
- 2. Grant, R.M.; Banden-Fuller, C.A. Knowledge accessing theory of strategic alliances. *J. Manag. Stud.* 2004, 41, 61. [CrossRef]
- 3. Mariti, P. Los acuerdos de colaboración entre empresas en las economías modernas: Algunos apuntes para la reflexión. *Econ. Ind.* **1989**, *1*, 61–67.
- 4. García Canal, E. Contractual form in Domestic and international strategic alliances. *Organ. Stud.* **1996**, 17, 773–794. [CrossRef]
- 5. Teece, D.J. Competition and cooperation: Striking the right balance. *Calif. Manag. Rev.* 1989, 31, 29.
- 6. Parkhe, A. Strategic alliance structuring: A game theoretic and transaction cost examination of interfirm cooperation. *Acad. Manag. J.* **1993**, *36*, 794–829.
- 7. Porter, M.E.; Fuller, M.B. Coaliciones y Estrategia Global. ICE 1988, 1, 101–120.
- 8. Speckman, R.E.; Kamauff, J.W., Jr.; Myhr, N. An empirical investigation into supply chain management. A perspective on partnerships. *Int. J. Phys. Distrib. Logist. Manag.* **1998**, *28*, 630–650. [CrossRef]
- 9. Glaister, K.W.; Buckley, J. Strategic motives for international alliance formation. *J. Manag. Stud.* **1996**, 33, 301–332. [CrossRef]
- 10. Saxton, T. The effects of partner and relationship characteristics on alliance outcomes. *Acad. Manag. J.* **1997**, 40, 443–461.
- 11. Pulgar, C.; Pelekais, C. Responsabilidad social basada en alianzas estratégicas en universidades de gestión privada. *Rev. Form. Gerenc.* **2014**, *13*, 199–220.
- Thorne, L.; Mahoney, L.S.; Gregory, K. A comparison of Canadian and US CSR Strategic alliances, CSR Reporting and CSR Performance: Insights into implicit-explicit CSR. *J. Bus. Ethics* 2017, 143, 85–98.
 [CrossRef]
- Van Der Voort, J.M.; Glac, K.; Meijs, L.C. 'Managing' corporate community involvement. J. Bus. Ethics. 2009, 90, 311–329. [CrossRef]
- 14. García, C.; Raufflet, E.; Portales, L. Las alianzas entre empresas y ONG como medio para promover la RSE y el desarrollo local. *Innov. RICEC* **2011**, *3*, 1–15.

- 15. Carreras, I.; Iglesias, M. Modelo de J. Austin Sobre Alianzas Sociales: La Colaboración Efectiva de las ONG; ESADE; Institute for Social Innovation Ramón LLull University: Barcelona, Spain, 2012. Available online: https: //www.pwc.es/es/fundacion/assets/pwc-esade-colaboracion-efectiva-ong.pdf (accessed on 8 September 2020).
- 16. Rodríguez, D.; Flores, R.; Miranda, P. Alianzas entre empresas lucrativas y organizaciones sin fines de lucro (OSFL): Estudios de casos en Chile. *Universum* **2013**, *28*, 173–202. [CrossRef]
- Abenoza, S.; Carreras, I.; Sureda, M. Colaboraciones ONG y Empresa que Transforman la Sociedad; Programa ESADE-PwC de Liderazgo Social 2014-15; Institute for Social Innovation Ramón LLull University: Barcelona, Spain, 2015. Available online: https://www.pwc.es/es/fundacion/assets/programa-esade-liderazgo-social-2015.pdf (accessed on 8 September 2020).
- 18. Porter, M.E.; Kramer, M.R. Creating shared value. How to Reinvent Capitalism—and Unleash a Wave of Innovation and Growth. *Harv. Bus. Rev.* **2011**, *89*, 62–77.
- 19. Dyllick, T.; Hockerts, K. Beyond de business case for corporate sustainability. *Bus. Strategy Environ.* 2002, *11*, 130–141. [CrossRef]
- Bansal, P. Evolving sustainability a longitudinal study of corporate sustainable development. *Strateg. Manag. J.* 2005, 26, 197–218. [CrossRef]
- 21. Schaltegger, S.; Wagner, M. Managing the Business Case for Sustainability: The Integration of Social, Environmental and Economic Performance, 1st ed.; Greenleaf Publishing: Sheffield, UK, 2006.
- 22. Eccles, R.G.; Miller-Perkins, K.; Serafiem, G. How to become a sustainable company. *MIT Sloan Manag. Rev.* **2012**, *53*, 43–50.
- 23. Montiel, I.; Delgado-Ceballos, J. Defining and measuring corporate sustainability: Are we there yet? *Organ. Environ.* **2014**, *27*, 113–139. [CrossRef]
- 24. Dyllick, T. Responsible management education for a sustainable world: The challenges for business schools. *J. Manag. Dev.* **2015**, *34*, 16–33. [CrossRef]
- 25. Dyllick, T.; Muff, K. Clarifying the meaning of sustainable business: Introducing a typology from business-as-usual to true business sustainability. *Organ. Environ.* **2016**, *29*, 156–174. [CrossRef]
- 26. Felber, C. Change Everything: Creating an Economy for the Common Good, 1st ed.; Zed Books: London, UK, 2015.
- 27. Felber, C.; Campos, V.; Sanchis, J.R. Common Good Balance Sheet, an Adequate Tool to Capture Non-Financials? *Sustainability* **2019**, *11*, 3791. [CrossRef]
- 28. Muñoz Martín, J.; Montoro Sánchez, M. Enfoques teóricos para el estudio de la cooperación empresarial. *Cuad. Estud. Empresariales* **2007**, *17*, 141–163.
- 29. Harrison, J.S.; Berman, S.L. Corporate Social Performance and Economic Cycles. J. Bus. Ethics 2016, 138, 279–294. [CrossRef]
- Canzaniello, A.; Hartmann, E.; Fifka, M.S. Intra-industry strategic alliances for managing sustainability related supplier risks motivation and outcome. *Int. J. Phys. Distrib. Logist. Manag.* 2017, 47, 387–409. [CrossRef]
- 31. Fusfeld, D.R. Joint Subsidiaries in the Iron and Steel Industry. Am. Econ. Rev. 1958, 48, 578–587.
- 32. Dixon, P.R. Joint Ventures: What is Their Impact on Competition? Antitrust Bull. 1962, 7, 397–410.
- 33. Mead, W.J. The Competitive Significance of Joint Ventures. *Antitrust Bull.* 1967, 12, 819–849.
- 34. Williamson, O.E. *The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting*, 1st ed.; Free Press: New York, NY, USA, 1985.
- Jacquemin, A.; Lammerant, M.; Spinoy, B. Compétition Européenne et Cooperation Entre Entreprises en Matière de Recherche-Développement, 1st ed.; Office des Publications Officielles des Communautés Européennes: Luxembourg, 1986.
- 36. Shapiro, C.; Willing, R.D. On the Antitrust Treatment of Production Joint Ventures. *J. Econ. Perspect.* **1990**, *4*, 113–130. [CrossRef]
- 37. Williamson, O.E. Transaction Cost Economics: The Governance of Contractual Relations. *J. Law Econ.* **1979**, 22, 233–261. [CrossRef]
- 38. Mariti, P.; Smiley, R.H. Co-operative Agreements and the Organization of Industry. J. Ind. Econ. 1983, 31, 437–451. [CrossRef]
- 39. Williamson, O.E. Comparative Economic Organization: The Analysis of Discrete Structural Alternatives. *Admin. Sci. Q.* **1991**, *36*, 269–296. [CrossRef]
- 40. Williamson, O.E. Markets, Hierarchies and the Modern Corporation: An Unfolding Perspective. *J. Econ. Behav. Organ.* **1992**, 17, 335–352. [CrossRef]

- 41. Evan, W.M. The Organization-Set: Toward a Theory of Interorganizational Relations. *Manag. Sci.* **1965**, 11, 217–230. [CrossRef]
- 42. Selznick, P. TVA and the Grass Roots, 1st ed.; University of California Press: Berkeley, CA, USA, 1949.
- 43. Levine, S.; White, P.E. Exchange as a Conceptual Framework for the Study of Interorganizational Relationships. *Admin. Sci. Q.* **1961**, *5*, 583–610. [CrossRef]
- 44. Benson, J.K. The Interorganizational Network as a Political Economy. *Admin. Sci. Q.* **1975**, *20*, 229–249. [CrossRef]
- 45. Harrigan, K.R. Joint Ventures and Global Strategies. Colum. J. World Bus. 1984, 19, 7–16.
- Hamel, G. Competition for Competence and Interpartner Learning Within International Strategic Alliances. Strateg. Manag. J. 1991, 12, 83–103. [CrossRef]
- 47. Contractor, F.; Lorange, P. Why should firm co-operate? The strategic and economic basis for co-operative ventures. In *Cooperative Strategies in International Business*, 1st ed.; Contractor, F., Lorange, P., Eds.; Lexington Books: Lexington, MA, USA, 1988; pp. 3–30.
- Lyons, T.; Krachenberg, A.; Henke, J. Mixed Motive Marriages: What's Next for Buyer-Supplier Relations? MIT Sloan Manag. Rev. 1990, 31, 29–36.
- 49. Harrigan, K.R. Strategies for Joint Ventures, 1st ed.; Lexington Books: Lexington, MA, USA, 1985.
- 50. Bleeke, J.; Ernst, D. The Way to Win in Cross-Border Alliances. Harv. Bus. Rev. 1991, 69, 127–135. [PubMed]
- 51. Das, T.K.; Teng, B. Sustaining Strategic Alliances: Options and Guidelines. J. Gen. Manag. 1997, 22, 49–64.
- 52. Koza, M.P.; Lewin, A.Y. The Co-evolution of Strategic Alliances. Organ. Sci. 1998, 9, 255–264. [CrossRef]
- 53. Kopka, K.; Mahoney, L.S.; Convery, S.P.; LaGore, W. An examination of alliances and corporate social responsibility. In *Research on Professional Responsibility and Ethics in Accounting*, 1st ed.; Jeffrey, C., Ed.; Emerald Group Publishing: Bingley, UK, 2014; Volume 18, pp. 109–130, ISBN 978-1-78441-164-0.
- 54. Ma, Y.Y.; Sia, C.L.; Li, Y.; Zheng, S. Sources of resources, alliance green management, and alliance performance in an emerging economy. *J. Clean. Prod.* **2016**, *139*, 319–327. [CrossRef]
- 55. Altamira, R. La negociación de alianzas estratégicas internacionales. Dir. Organ. 2000, 24, 99–106.
- 56. Michavila, F. Del aislacionismo a las alianzas. Arbor 2011, 187, 67–75. [CrossRef]
- 57. Renart Cava, L.G. Consorcios de exportación y otros tipos de alianzas estratégicas entre empresas. *Bol. Econ. ICE* **1999**, *1*, 1001–1016.
- 58. Pérez López, J.A. Fundamentos de la Dirección de Empresas, 7th ed.; Ediciones Rialp: Madrid, Spain, 1993.
- 59. Browning, L.; Beyer, J.M.; Shetler, J.C. Building cooperation in a competitive industry: Sematech and the semiconductor industry. *Acad. Manag. J.* **1995**, *38*, 113–151.
- 60. Amran, A.; Lee, S.P.; Devi, S.S. The influence of governance structure and strategic corporate social responsibility toward sustainability reporting quality. *Bus. Strategy Environ.* **2014**, *23*, 217–235. [CrossRef]
- 61. Lai, J.H.; Chen, L.Y. The valuation effect of corporate governance on stakeholder wealth: Evidence from strategic alliances. *Int. Rev. Econ. Finance* **2014**, *1*, 117–131. [CrossRef]
- 62. Silvestri, A.; Veltri, S. Ethical strategic alliances for the sustainable recovering of a territory, Corporate Governance. *Int. J. Bus. Soc.* **2017**, *17*, 230–249.
- 63. Kliksberg, B. *Ética Empresarial: ¿Moda o Demanda Imparable?* Formación Carolina Universidad de Buenos Aires: Buenos Aires, Argentina, 2006. Available online: http://cyta.com.ar/biblioteca/bddoc/bdlibros/rse/334_as_etica_moda.pdf (accessed on 8 September 2020).
- 64. Rueda López, R. La Ética Como Fundamento de la Responsabilidad Social: El Fomento de la Perspectiva de Género Como Praxis Socialmente Responsable en la Universidad. Ph.D. Thesis, University of Córdoba, Córdoba, Spain, 2016.
- 65. Morales, H.N.; González, J.G.; Torres, M.A. La Creación de Valor Compartido y la Innovación Social Como Factores de Sinergia Para el Desarrollo Económico; Instituto de Ciencias Básicas e Ingeniería UAEH: Hidalgo, Mexico, 2011. Available online: http://acacia.org.mx/busqueda/pdf/11_PF721_Valor_Compartido_e_Innovaci_n_Social.pdf (accessed on 8 September 2020).
- Berger, I.E.; Cunningham, P.H.; Drumwright, M.E. Social alliances: Company/nonprofit collaboration. *Calif. Manag. Rev.* 2004, 47, 58–65. [CrossRef]
- 67. Arya, B.; Salk, J.E. Cross-sector alliance learning and effectiveness of voluntary codes of corporate social responsibility. *Bus. Ethics Q.* **2006**, *16*, 211–234. [CrossRef]
- Rivera-Santos, M.; Rufin, C. Odd couples: Understanding the governance of firm-NGO alliances. *J. Bus. Ethics* 2010, 94, 55–70. [CrossRef]

- Shumate, M.; O'Connor, A. Corporate reporting of cross-sector alliances: The portfolio of NGO partners communicated on corporate websites. *Commun. Monogr.* 2010, 77, 207–230. [CrossRef]
- 70. Hansen, E.G.; Sextl, M.; Reichwald, R. Managing strategic alliances through a community-enabled Balanced Scorecard: The case of Merck Ltd., Thailand. *Bus. Strategy Environ.* **2010**, *19*, 387–399. [CrossRef]
- 71. Liu, G.; Ko, W.W. An analysis of cause-related marketing implementation strategies through social alliance: Partnership conditions and strategic objectives. *J. Bus. Ethics* **2011**, *100*, 253–281. [CrossRef]
- 72. Im, J. The role of NGO partner's knowledge and reputation on corporate learning in cross-sector alliances. *J. Strateg. Manag.* **2011**, *14*, 67–88.
- 73. Al-Tabbaa, O.; Leach, D.; March, J. Collaboration between nonprofit and business sectors: A framework to guide strategy development for nonprofit organizations. *Voluntas* **2014**, *25*, 657–678. [CrossRef]
- 74. Rim, H.; Yang, S.U.; Lee, J. Strategic partnerships with nonprofits in corporate social responsibility CSR: The mediating role of perceived altruism and organizational identification. *J. Bus. Res.* **2016**, *69*, 3213–3219. [CrossRef]
- Muthuri, J.N.; Chapple, W.; Moon, J. An Integrated Approach to Implementing Community Participation in Corporate Community Involvement: Lessons from Magadi Soda Company in Kenya. J. Bus. Ethics 2009, 85, 431–444. [CrossRef]
- 76. Radrigán, M.H.; Dávila, A.M.; Penaglia Vasquez, F.E. Gestión y liderazgos en los emprendimientos sociales: El caso del sector no lucrativo chileno, Polis. *Rev. Latinoam.* **2012**, *11*, 1–15.
- 77. Callejas, R.J.M.; Salazar, J. La Alianza Universidad-Empresa-Estado y los procesos de I+ D: Una mirada al caso antioqueño. *Perf. Coyunt. Econ.* **2012**, *1*, 157–176.
- 78. Romero, A.M.; Oliva, E.J.D. Formas y categorización de las relaciones entre las organizaciones sin fines de lucro y el sector privado: Una revisión. *Adm. Desarro.* **2012**, *40*, 77–92. [CrossRef]
- 79. Brandenburger, A.M.; Nalebuff, B.J. Co-Opetition, 1st ed.; Harvard Business School: Boston, MA, USA, 1996.
- Bengtsson, M.; Eriksson, J.; Wincent, J. Coopetition: New ideas for a new paradigm. In *Coopetition: Winning* Strategies for the 21st Century, 1st ed.; Edward Elgar Publishing: Cheltenham, UK, 2010; pp. 19–39.
- Bouncken, R.B.; Fredrich, V. Coopetition: Performance implications and management antecedents. *Int. J. Innov. Manag.* 2012, 16, 1–28. [CrossRef]
- 82. Rusko, R. Perspectives on value creation and coopetition. Prob. Perspect. Manag. 2012, 10, 60–72.
- 83. Cygler, J.; Gajdzik, B.; Sroka, W. Coopetition as a development stimulator of enterprises in the networked steel sector. *Metalurgija* **2014**, *53*, 383–386.
- 84. Reaidy, P.J.; Gunasekaran, A.; Spalanzani, A. Bottom-up approach based on Internet of Things for order fulfillment in a collaborative warehousing environment. *Int. J. Prod. Econ.* **2015**, 159, 29–40. [CrossRef]
- 85. Campos, V.; Sanchis, J.R.; Ejarque, A. Social entrepreneurship and Economy for the Common Good: Study of their relationship through a bibliometric analysis. *Int. J. Entrep. Innov.* **2020**, *21*, 156–167. [CrossRef]
- 86. Economy for the Common Good Companies. Available online: https://www.ecogood.org/apply-ecg/ companies/#balance-sheet-resources (accessed on 10 September 2020).
- 87. Yin, R.K. Validity and generalization in future case study evaluations. Evaluation 2013, 19, 321–332. [CrossRef]
- Rule, P.; John, V.M. A necessary dialogue: Theory in case study research. *Int. J. Qual. Methods* 2015, 14, 1–11. [CrossRef]
- 89. Yin, R.K. Case Study Research: Design Methods, 3rd ed.; SAGE Publications: Thousand Oaks, CA, USA, 2002.
- Planellas, M.; Urriolagoitia, L. La Alianza Estratégica Entre Grupo Vips y Fundación Hazlo Posible. Un Ejemplo de Colaboración Empresa-ONG; ESADE, Institute for Social Innovation, Ramón LLull University: Barcelona, Spain, 2011. Available online: http://itemsweb.esade.es/wi/research/iis/publicacions/10906_Alianza_Estrategica.pdf (accessed on 10 September 2020).
- 91. Grupo Vips. Hacemos Comunidad. El Voluntariado Corporativo es Parte de Nuestro ADN. Available online: https://www.grupovips.com/quienes-somos/responsabilidad-corporativa/hacemos-comunidad (accessed on 10 September 2020).
- 92. Talento Que Impacta. Available online: https://www.hazloposible.org/wp-content/uploads/2019/03/Red-Talento-que-Impacta-Empresas-2019.pdf (accessed on 10 September 2020).
- 93. Alianzas y Colaboraciones. Available online: https://www.hazloposible.org/alianzas-y-colaboraciones/ (accessed on 10 September 2020).
- 94. Grupo Vips Datos Clave. Available online: https://www.grupovips.com/quienes-somos (accessed on 10 September 2020).

- 95. Danone: One Planet, One Health. Available online: https://www.danone.com/about-danone/Danone-yogurtturns-100/looking-back-at-an-epic-journey.html (accessed on 10 September 2020).
- 96. Introduction Grameen Bank. Available online: http://www.grameen.com/introduction/ (accessed on 10 September 2020).
- 97. Fisac, R.; Moreno, A.; Mataix, C.; Palacios, M. *Emprendimiento Social*; Revista Española del Tercer Sector; Fundación Luis Vives: Madrid, Spain, 2011. Available online: http://www.plataformatercersector.es/sites/ default/files/N17%20RETS%20Emprendimiento%20social.pdf (accessed on 10 September 2020).
- 98. Barrera, E. La empresa social y su responsabilidad social. *Innovar* **2007**, *17*, 59–76.
- 99. Danone Communities: Our story. Available online: http://www.danonecommunities.com/index.php/ alleviate-poverty-fr/ (accessed on 10 September 2020).
- 100. Danone Communities: What We Do. Available online: http://www.danonecommunities.com/ (accessed on 10 September 2020).
- 101. Grameen—DANONE Foods Limited. Available online: https://www.fundacionseres.org/Publicaciones/ proyectos-colaboracion-entidad/grameen-danone-foods-limited/ (accessed on 10 September 2020).
- 102. Reptrak: What is Your Company Reputation? Available online: https://www.reptrak.com/blog/ (accessed on 10 September 2020).
- 103. Grameen Foundation Our Family of Organizations. Available online: https://grameenfoundation.org/about-us/the-grameen-family (accessed on 10 September 2020).
- Sanchis, J.R.; Campos, V.; Ejarque, A. La Economía del Bien Común como modelo transformador. Análisis comparativo por países en Europa. *REM* 2020, 54, 87–106.

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).